

Impacts of Changing Atmospheric Circulation Patterns on Aviation Turbulence Over Europe

Tommaso Alberti, Davide Faranda, Lia Rapella, Erika Coppola, Fabio

Lepreti, Bérengère Dubrulle, Vincenzo Carbone

► To cite this version:

Tommaso Alberti, Davide Faranda, Lia Rapella, Erika Coppola, Fabio Lepreti, et al.. Impacts of Changing Atmospheric Circulation Patterns on Aviation Turbulence Over Europe. Geophysical Research Letters, 2024, 51, pp.2024GL111618. 10.1029/2024GL111618. insu-04879848

HAL Id: insu-04879848 https://insu.hal.science/insu-04879848v1

Submitted on 10 Jan 2025 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2024GL111618

Key Points:

- Changing atmospheric circulation due to climate change increases turbulence over Europe
- Turbulence peaks in winter, linked to wind shears from the subtropical jet stream over the Southern Mediterranean
- Most increasing episodes are related to clear air turbulence, occurring unexpectedly at flight cruise altitudes

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

T. Alberti, tommaso.alberti@ingv.it

Citation:

Alberti, T., Faranda, D., Rapella, L., Coppola, E., Lepreti, F., Dubrulle, B., & Carbone, V. (2024). Impacts of changing atmospheric circulation patterns on aviation turbulence over Europe. *Geophysical Research Letters*, *51*, e2024GL111618. https://doi.org/10.1029/ 2024GL111618

Received 30 JUL 2024 Accepted 11 NOV 2024

© 2024. The Author(s). This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is

properly cited.

Impacts of Changing Atmospheric Circulation Patterns on Aviation Turbulence Over Europe

Tommaso Alberti¹ ^(D), Davide Faranda^{2,3,4} ^(D), Lia Rapella^{2,5}, Erika Coppola⁶ ^(D), Fabio Lepreti⁷ ^(D), Bérengère Dubrulle⁸ ^(D), and Vincenzo Carbone⁷ ^(D)

¹Istituto Nazionale di Geofisica e Vulcanologia, Rome, Italy, ²Laboratoire des Sciences du Climat et de l'Environnement, Gif-sur-Yvette, France, ³London Mathematical Laboratory, London, UK, ⁴Laboratoire de Météorologie Dynamique/IPSL, École Normale Supérieure, PSL Research University, Sorbonne Université, Paris, France, ⁵LMD-IPSL, Ecole Polytechnique, CNRS, Palaiseau, France, ⁶Abdus Salam ICTP, Trieste, Italy, ⁷Università della Calabria, Dipartimento di Fisica, Rende, Italy, ⁸CEA, CNRS, Gif-sur-Yvette, France

Abstract Addressing aviation turbulence not only enhances passenger comfort but also help to reduce fuel consumption and environmental impact, supporting aviation sustainability. By using ERA5 reanalysis data we explore how changing atmospheric circulation, monitored via sea level pressure trends, impacts aviation turbulence over Europe. Our results show coherent climate anomalies, with rising turbulence intensity over the UK and Northern Europe where most events involve clear air turbulence, which occurs unexpectedly and without warning, particularly at flight altitudes. We also highlight a clear seasonal patterns in moderate-or-greater turbulence encounters, most frequent and intense in winter, with a key role of wind shears due to the sub-tropical jet position over Southern Mediterranean. Our approach adds to previous studies on the same topic by analyzing individual atmospheric circulation pattern changes and their effects on turbulence-related factors. This offers insight into how climate change affecting atmospheric dynamics may contribute to increased aviation turbulence.

Plain Language Summary This study investigates how climate change can affects aviation turbulence over Europe, which is crucial for improving passenger comfort, reducing fuel consumption, and promoting sustainable aviation. Our findings indicate an increase in turbulence, particularly over the UK and Northern Europe with a significant portion of this increase involving clear air turbulence. This is particularly hazardous as it occurs without warning and cannot be detected visually, especially at cruising altitudes. We also found that turbulence is more frequent and intense during winter with a seasonal pattern influenced by wind shears associated with the position of the subtropical jet stream over the Southern Mediterranean. Finding ways to face with turbulence could make flights smoother, safer, and more environmentally friendly.

1. Introduction

The unprecedented changes in Earth's climate are reshaping atmospheric dynamics on a global scale (Masson-Delmotte et al., 2021), with profound implications for various sectors, including aviation (Shaw & Miyawaki, 2023; Williams & Joshi, 2013). Atmospheric patterns changes modify the weather and, in synergy with global warming, enhance extreme events (e.g., storms, precipitations, heatwaves, droughts, Christidis and Stott (2021); Gudmundsson et al. (2017); Bevacqua et al. (2019)), affecting different sectors (e.g., industry, agriculture, tourism, Ortiz-Bobea et al. (2021); Scott et al. (2012)). As anthropogenic activities continue to drive climate change, understanding the specific regional manifestations of these alterations becomes imperative, particularly over Europe which is marked by densely populated urban centers, major economic activities, and extensive air traffic networks (D. S. Lee et al., 2009). Indeed, one of the most evident consequences of global warming is the intensification of convective phenomena (Seneviratne et al., 2021). As temperatures increase, also the moisture content in the atmosphere increases, leading to convective processes and fostering the development of thunderstorms and intense rainfall events (Dayan et al., 2015). These convective systems can also generate strong wind shears, contributing to atmospheric instability and turbulence, particularly in the lower atmosphere where most commercial aircraft operate (Sharman & Lane, 2016).

European climate features a rich meteorology, with its diverse geography, orography, and micro-climates, which is greatly impacted by climate change (Naumann et al., 2021). Of particular interest is the heightened potential for increased aviation turbulence, with direct consequences for flight operations, fuel efficiency, passenger comfort,

and overall safety (Gratton et al., 2022). Indeed, atmospheric turbulence poses a significant risk to the aviation industry, accounting for 71% of all weather-related aircraft accidents among regularly scheduled air carriers (Gultepe & Feltz, 2019). Among different types (Storer et al., 2019; J.-H. Kim & Chun, 2010; Lane et al., 2012; Trier et al., 2022) clear-air turbulence (CAT), defined as high-altitude aircraft bumpiness occurring away from significant cloudiness and thunderstorm activity, is, surprisingly, the half of all encounters with turbulence (Williams & Storer, 2022). Recent studies (Storer et al., 2017; Williams, 2017; Williams & Joshi, 2013) have shown a significant increase in turbulence levels over the last decades which are projected to further increases according to climate models (Smith et al., 2023) by mid-century in various regions, including the North Atlantic, North America, and Europe (Wolff & Sharman, 2008; Storer et al., 2017; S. H. Lee et al., 2019). Given the wide spectrum of aviation-related weather impacts and the inherent challenge in accurately simulating them through a singular or a set of models (as global climate models Smith et al. (2023)), a holistic approach is helpful to discern increasing or decreasing trends in the frequency of occurrence of atmospheric circulation patterns (Faranda, Messori, et al., 2023a; Faranda et al., 2023b). Indeed, the interconnected nature of meteorological events across the European aviation sector requires for an integrated analysis, in which different atmospheric observables are jointly considered to account for both phenomena and impacts.

In this study, by using the ERA5 reanalysis data (Hersbach et al., 2023) we investigate the extent to which changes in atmospheric dynamics amplify the impacts of climate change on aviation turbulence at regional scales (over North Atlantic and European sectors). Our analysis centers on the identification of weather patterns derived from the classification of daily mean sea level pressure (SLP) field (Faranda, Messori, et al., 2023a; Faranda et al., 2023b). This atmospheric variable is a good tracker of the atmospheric dynamics, at mid-latitude, and it is the most used variable for attribution studies (F. E. L. Otto, 2016; F. Otto, 2019), justifying its historical prominence in meteorological broadcasts, reanalysis products, and models of different complexity (Hersbach et al., 2023).

By focusing on the morphology of large-scale SLP patterns that are more frequently encountered during the last 4 decades in the ERA5 data (Hersbach et al., 2023) we can devise the most-encountered spatial pattern of key diagnostics (e.g., Ellrod's indices TI1 and TI2) related to aviation turbulence episodes. We observe a growing frequency of large-scale atmospheric patterns conducive to moderate-or-greater (MOG)-level turbulence over the UK and Northern Europe. Our approach complements to previous studies (e.g., Williams and Storer (2022); Storer et al. (2017); Shaw and Miyawaki (2023)) in examining long-term changing trends in atmospheric circulation patterns and their impacts on turbulence-related variables to provide a holistic understanding of how variations in atmospheric dynamics contribute to the intensification of aviation turbulence.

2. Data and Methods

2.1. Data

We use the most recent climate reanalysis data generated by the European Centre for Medium-Range Weather Forecasts (ECMWF) as a part of the EU-funded Copernicus Climate Change Service (C3S). Specifically, our focus was on the ERA5 data set, providing hourly sampled atmospheric parameters spanning from 1950 to the present, with a horizontal resolution over Europe of $0.25^{\circ} \times 0.25^{\circ}$, and at different pressure levels (Hersbach et al., 2023). We decided to focus on the satellite era (i.e., 1979-today) when widespread observations of climate variables from satellites have become available. This consistent coverage over an extended 44-year period enables us to detect temporal trends in the dynamics (from hours up to inter-annual/decadal timescales) associated with spatial patterns, also ensuring uniform spatial and temporal coverage. We focus over the European box extending between 30°W and 40°E in longitude and between 35°N and 75°N in latitude with a grid size of $281 \times$ 181 cells (Hersbach et al., 2023). We use mean sea level pressure as a key indicator for analogs searching (e.g. (Faranda, Messori, et al., 2023a; Faranda et al., 2023b),), while for investigating impacts on aviation turbulence we use three different parameters: the Ellrod's indices TI1 and TI2 and the eddy dissipation rate (EDR). The Ellrod's indices, TI1 and TI2, are turbulence indices developed to help forecast CAT in the atmosphere. TI1 (in units of s^{-1}) is based on the deformation of the wind field at flight levels by combining the vertical wind shear (VWS) and the horizontal deformation (DEF), that is, $TI1 = VWS \times DEF$. High values of TI1 suggest a high potential for CAT, primarily associated with wind shear zones such as jet streams (Ellrod & Knapp, 1992). TI2 (in units of s^{-1}) further incorporates convergent flow (-DIV) related to the development of frontal zones and thereby provides a more refined estimate of turbulence by considering both dynamic and thermodynamic properties of the

atmosphere (Ellrod & Knox, 2010), that is, TI2 = TI1-VWS × DIV. Like their TI1 counterpart, high values of TI2 are related to more likely CAT encounters. Finally, the eddy dissipation rate EDR (in units of $m^{2/3} s^{-1}$) represents the rate at which turbulent kinetic energy is dissipated per unit mass due to the effects of turbulent eddies ((ICAO), 2018). We indirectly derived the EDR from the cube-root of the kinetic energy estimated via the velocity field provided by the ERA5 database (Hersbach et al., 2023). Indeed, the European Center for Medium-Range Weather Forecasts (ECMWF) is being to release EDR as a product in the new reanalysis version ERA6 (Bechtold et al., 2021). For our purposes, since we cannot directly access to the inertial range of turbulence due to the standard outputs of ERA5 reanalysis, we estimate the EDR by assuming valid the Kolmogorov's second hypothesis of similarity and by employing the Taylor's hypothesis is used to express spatial increments as temporal ones. Thus,

$$EDR = \left[\frac{V^2}{\Delta t}\right]^{1/3} \tag{1}$$

where $V' = \sqrt{\langle V^2 \rangle}$ is the root-mean square (RMS) of the velocity field and Δt is the time resolution. This is equivalent to the second-order structure function approach (Frehlich & Sharman, 2004; Kim et al., 2022) but adapted to the outputs provided by ERA5. We need to point out that our estimation can be considered as a proxy of the EDR usually adjusted for aircraft type. Indeed, our EDR is not derived from AMDAR transmissions and it is not adapted on the type of aircraft ((ICAO), 2018). However, its statistics (see Figure S6 in Supporting Information S1) is well in agreement with values reported in literature, allowing to classify aviation turbulence onto different classes/levels ((ICAO), 2018): null (EDR lower than 0.1), light (EDR between 0.1 and 0.2), moderate (EDR between 0.2 and 0.45), and severe (EDR larger than 0.45).

For our purpose we focus on the range of altitudes between 250 hPa (~34 kft) and 200 hPa (~38 kft) which includes cruising altitudes for commercial airliners (~36–37 kft) (Sharman & Lane, 2016).

2.2. Methods: Occurrence Trends of Analogs

The computation of the trends in the occurrence of atmospheric circulation patterns involves the following steps (Faranda, Messori, et al., 2023a; Faranda et al., 2023b). First of all, anomalies of the sea-level pressure field are obtained by removing a grid-by-grid point linear trend throughout the entire analysis period, to account for long-term thermodynamic trends, and a mean seasonal cycle by averaging over the corresponding calendar days. Then, Euclidean distances between these maps at different times are computed, treating one map as a reference state and measuring its distance from all others in the data set. A quantile (typically q = 0.98) determines which fields are considered analogs. The entire time interval is further divided into equal sub-intervals, and the number of analogs N(t) within each sub-interval is counted. A linear best-fit detrending process is applied as

$$N(t) = a t + b \tag{2}$$

and the 95% confidence intervals (CI) for the slope of the fit (*a*) are calculated. If both the lower and upper CI bounds are positive or negative, then that SLP pattern is associated with a positive or negative trend. If the CI includes zero, the trend is considered non-significant. The robustness of the estimation is assessed by both repeating calculations for higher quantiles (q = 0.99, 0.995), as well as, by bootstrapping procedures (Faranda, Messori, et al., 2023a; Faranda et al., 2023b). Thus, we consider only maps displaying consistent and significant occurrence trends across all quantiles as robust circulation patterns.

Finally, since the output of the analogs searching is a list of dates they can be used to reconstruct composite maps of the SLP field or of any other field of interest (e.g., temperature, wind speed, precipitation, turbulence indices, etc.), as well as, to inspect seasonal variations in the occurrence of analogs. The detection of occurrence trends of atmospheric analogs has proved useful impactful climate extremes (Jézéquel, Cattiaux, et al., 2018; Jézéquel, Dépoues, et al., 2018; Platzer et al., 2021; Yiou et al., 2017), including heatwaves and windstorms (Faranda, Messori, et al., 2023a; Faranda et al., 2023b), Acqua Alta events in Venice (Alberti et al., 2023; Faranda et al., 2023), droughts (Faranda, Pascale, & Bulut, 2023), and off-shore wind power availability (Rapella et al., 2023). In the following we specifically focus on increasing trends, that is, with coefficients *a* which are positively defined within error bars, to investigate the morphology of the patterns which are expected to be most



Geophysical Research Letters



Figure 1. Composite anomalies of sea-level pressure (a) and monthly percentage distribution (b) for days with increasing occurrence trends.

encountered and whose increases can be related to climate change (Faranda, Messori, et al., 2023a; Faranda et al., 2023b).

3. Results and Discussion

When averaging over all seasons, the atmospheric circulation patterns (Figure 1) with occurrence trends show highly coherent surface climate anomalies over Europe. Specifically, more frequently occurring patterns (Figure 1a) show a north–south cyclonic–anticyclonic bipolar configuration with anomalies ranging between ± 5 hPa. Increasing analogs patterns are also associated with anomalously windy, warm, and wet conditions in Northern and Eastern Europe and moderately warm, dry conditions in Southern Europe (Faranda, Messori, et al., 2023a; Faranda et al., 2023b). Furthermore, increasing trends are mostly occurring (Figure 1b) during winter season (DJF, ~50%), also showing a pronounced monthly variability, with occurrence reducing from 15% to 20% to less than 5% when moving from winter to summer months.

While previous studies (e.g. (Faranda, Messori, et al., 2023a; Faranda et al., 2023b; Rapella et al., 2023),) focused on meteorological variables (e.g., temperature, precipitation, wind speed) to study impacts of atmospheric circulation changes, here we are interested in variables which are strictly related with the occurrence of moderate-or-greater aviation turbulence (Sharman & Lane, 2016). As in J. H. Lee et al. (2023) we select MOG thresholds from the 95th percentiles of turbulence indices TI1, TI2, and EDR at 200 hPa. As a more conservative choice we evaluate this threshold only for increasing trends and not for the whole period (1979-today). The derived thresholds are in agreement with J. H. Lee et al. (2023), although slightly lower than Prosser et al. (2023). Nevertheless, our results are robust with respect to the selected thresholds for MOG turbulence. Figure 2 reports the composite maps for MOG turbulence of TI1 (a), TI2 (b), and EDR (c) for more frequently occurring patterns.

We can evidence that jointly larger TI1 and TI2 values are observed over the UK and the Norway, which we can linked to the presence of larger vertical wind shears (see Figure S1a in Supporting Information S1), while larger TI1 and TI2 values over the North Atlantic are mainly related to both vertical wind shears and larger values for deformation. The latter, being associated with frontogenesis (J. H. Lee et al., 2023), seems to play a dominant role also over Alps and Southern Mediterranean (see Figure S1b in Supporting Information S1), which can be mainly related to thermal gradients leading to shears (Sharman & Lane, 2016). Furthermore, larger values of TI2 are also observed over the Northern France, Northern Germany, as well as, over the Alps, which we can related to





Figure 2. Composite maps averaged between 250 hPa and 200 hPa of TI1 (a), TI2 (b), and EDR (c) for more frequently occurring patterns corresponding to MOG turbulence.

development of upper-level frontal zones (TI2 indeed also consider the divergence as an additional term with TI1, see Figure S1c in Supporting Information S1). Finally, larger EDR values are found in close correspondence of the south-east part of the European domain, close to the subtropical jet position, while lower values are encountered over the North Atlantic, UK, and Scandinavian regions.

By further inspecting the seasonal behavior of the observed patterns we can see that during winter months (Figure 3) the spatial extension as well as also the strength of turbulence is generally higher with respect to summer months (Figure 4).

In detail, we can see that during winter (Figure 3) T11 and T12 display similar spatial patterns with the composite maps (Figure 2), although with slightly overall larger values, as well as, also with localized differences. As an example, over Southern Mediterranean and Northern Africa we observe larger values for T11 that could be linked with frontogenesis (see Figures S2b in Supporting Information S1) located on the cyclonic shear side of the sub-tropical jet stream (Ellrod & Knapp, 1992). The larger values can be indeed explained by the generally stronger jet in the winter due to the more pronounced temperature gradient between the equator and the poles. Furthermore, the presence of cold air masses distinct from warm air ones creates favorable conditions for a steeper pressure gradient that strengthens the jet, leading to more frequent and intense weather systems often bringing wet and stormy conditions over Southern Mediterranean. Additionally, stronger MOG episodes can be detected over the Alps which can be related to the presence of jet streaks, more common in winter, and able in creating localized



Figure 3. Composite maps averaged between 250 hPa and 200 hPa of TI1 (a), TI2 (b), and EDR (c) for more frequently occurring patterns corresponding to MOG turbulence during winter.





Figure 4. Composite maps averaged between 250 hPa and 200 hPa of TI1 (a), TI2 (b), and EDR (c) for more frequently occurring patterns corresponding to MOG turbulence during summer.

areas of intense wind shear and horizontal deformation, further increasing the TI1 index. Overall, by looking the spatial extension of TI2 associated with MOG turbulence does not change in winter as compared with the overall seasons, although we can clearly evidence larger values which can likely produce stronger turbulence. Indeed, by further inspecting the EDR we evidence larger values in winter, meaning stronger MOG turbulence levels when flying within Europe.

Conversely, during summer the spatial extension of MOG turbulence significantly reduces (Figure 4), as well as the observed values for all metrics, with a significant reduction especially for TI2. Small-scale localized increases in TI1 are only observed again over the North Atlantic, related to deformation likely developing on the shear side of cyclonic jet stream (see Figure S3b in Supporting Information S1) and over the North Sea, between UK and Scandinavian region. We do not evidence MOG turbulence over Southern Mediterranean region, although in the summer, the subtropical jet stream shifts northward. Indeed, its strength is typically weaker in the summer with southern parts of Europe, such as the Mediterranean region, experiencing drier and more stable weather, with effects usually less intense than in the winter. TI2, instead, showing a northward extension of localized larger values toward Sweden and Finland, likely linked to slight increases in vertical wind shears (see Figure S3a in Supporting Information S1). Finally, also the EDR evidence generally lower values and low-extended regions for MOG encounters. However, we need to point out that, given the use of CAT diagnostics (TI1 and TI2), the occurrence of CIT during summer could be missed (e.g., J.-H. Kim and Chun (2011); S.-H. Kim et al. (2021)), although we clearly identify a different role of VWS and DEF in leading TI1 patterns between summer and winter.

Similar scenarios to winter, although with lower values for all turbulence-related metrics, are also found for the European spring (see Figure S4 in Supporting Information S1), with also higher values for TI1 over Central and Southern Mediterranean, where it is expected to observe wind shears likely related to the increased convective activity, including thunderstorms, which can generate localized wind shear. Concerning EDR, we observe lower values over the northern part of the domain, while similar (high) values are observed for the south-eastern part, likely to be attributed to the sub-tropical jet stream. Conversely, particularly for the TI1 index, autumn season (see Figure S5 in Supporting Information S1) is quite similar to the summer time, although with larger extension of regions where TI2 assumes high values. Finally, EDR is mostly characterized by values close to 0.22, with a similar spatial extent of the winter and spring seasons. The different behavior over seasons compliments a recent study by Prosser et al. (2023) who found a larger absolute increase in winter times with respect to summer ones for MOG turbulence, as well as, also an increase of the time spent in MOG conditions.

Overall, by looking at the statistics for the EDR (see Figure S6 in Supporting Information S1) we depict an overall increase in its values for more-frequently encountered patterns with respect to the average, together with an increase in the fraction of MOG turbulence likely due to the increased role of climate change of the last 4 decades (Faranda, Messori, et al., 2023a; Faranda et al., 2023b).





Figure 5. Composite maps of the probability of occurrence of MOG turbulence based on TI1 (a), TI2 (b), and EDR (c) for more frequently occurring patterns.

As a further step, the main question to address is where it is more probable to encounter MOG turbulence over the European domain, as well as, whether increasing or decreasing trends are also associated with increased turbulence strengths. Thus, we evaluate the fraction of time spent above the MOG threshold evaluated as the 95th percentile over the full record (1979–2023, not only on the subset of positively increasing trends) as reported in Figure 5.

The most striking feature that emerges is a significant higher chance to encounter MOG turbulence episodes close to the subtropical jet position. This suggests that the subtropical jet stream is still the main pattern enhancing turbulence episodes over the Southern regions of the European airspace; however, its role has been enhanced, in terms of probability of occurrence and intensity, due to long-term climate trends. Indeed, the probability of encountering MOG turbulence due to the subtropical jet stream moves from about 1.5% during decreasing trends up to 4% for increasing trends. This finding on the role of enhanced jet streams in leading more intense turbulence is in agreement with previous works on this topic (Prosser et al., 2023; Shaw & Miyawaki, 2023; Williams & Joshi, 2013). However, the main result is that there is a complimentary behavior for increasing trends when focusing over Europe for encountering MOG turbulence. While increasing levels have been previously reported over the North Atlantic (Williams & Joshi, 2013), here we show that also flying within the European airspace (e.g., toward central-to-northern regions or crossing southern Mediterranean during winter) can lead to more MOG turbulence episodes. This MOG turbulence tends to occur more frequently over a larger spatial area, extending from North Atlantic up to Scandinavian regions, or toward Central and Southern Mediterranean (depending on the season).

4. Conclusions

Our analysis reveals significant shifts in atmospheric circulation patterns over Europe and their implications for aviation turbulence. Consistent with prior studies (Faranda, Messori, et al., 2023a; Faranda et al., 2023b), we observe that increasing analog patterns correlate with windy, warm, and wet conditions in Northern and Eastern Europe. Furthermore, our analysis reveals a pronounced change in the seasonality of analog occurrence, indicative of an underlying trend attributed to climate change. Increasing trends predominantly occur during winter months, suggesting a notable shift in weather patterns.

Importantly, our study focuses on dynamic variables directly related to aviation turbulence, providing insights beyond traditional meteorological variables (Faranda, Messori, et al., 2023a; Faranda et al., 2023b). We observe distinct patterns in turbulence indicators as T11, T12, and EDR, indicative of evolving turbulence conditions as a consequence of varying atmospheric circulation patterns. Our analysis suggests that occurrence of MOG turbulence over the UK, Norway, and the North Atlantic can be linked to vertical wind shears and deformation. Conversely, in Southern Mediterranean MOG turbulence is associated with thermal gradients and wind shears, with the subtropical jet playing a significant role. MOG turbulence is more intense in winter, with larger values of T11 and T12, particularly over the Southern Mediterranean and Alps, due to stronger jet streams and weather systems. In summer, MOG turbulence is reduced, with smaller areas affected, mainly over the North Atlantic and

19448007, 2024, 23, Downlo

com/doi/10.1029/2024GL111618 by Cochrane

Wiley Online Library on [10/01/2025]. See the Terms

and Conditi

lley Online

Library for rules of use; OA

applicable Creative

North Sea. Our results indicate that the subtropical jet enhances turbulence, especially in Southern Europe, and long-term climate trends indicate an increasing likelihood (up to 4%) of encountering MOG turbulence due to stronger jet streams. Over the past four decades, turbulence intensity and frequency have risen, particularly in winter over the North Atlantic, Europe, and the Mediterranean.

Our study also highlights the potentials of the use of analogs to provide a different but complimentary approach with respect to the existing literature on the link between climate change and turbulence impacts on aviation (e.g., Williams (2017); Williams and Joshi (2013); Williams and Storer (2022); J. H. Lee et al. (2023)). Indeed, previous works focused on a range of parameters (up to 21) which can be directly related to turbulence. Then, to investigate the role of climate change, both using reanalysis data and climate model simulations, a temporal trend is extracted from each grid box during a selected time interval of interests (for historical records typically from the beginning of satellite observations, i.e., 1979). Thus, long-term trend variability is inspected as a linear trend over certain regions. Our approach rests in a different way that can be used for a single event or a set of similar (analog) events, and it is the core of the ClimaMeter framework (Faranda et al., 2023). It is based on looking for weather conditions similar to those that caused the extreme event of interest with statistical algorithms based on dynamical system metrics. The object studied is a surface-pressure pattern over a certain region and the estimation of trends is based on the number of similar patterns (analogs) to each of the considered one in the data set. Then, once we have dates for increasing occurrence trends we explore composite maps of turbulence-related variables. Thus, we use SLP field to diagnose atmospheric circulation changes due to climate change and then hazard variables (TI1, TI2, and EDR) to explore effects on turbulence-related metrics. Finally, beyond the set of variables that can be used to assess turbulence strength (Williams & Joshi, 2013) we focus on these three one since they can provide easy metrics to directly determine turbulence levels as well as the main physical source mechanisms of the observed episode (CAT, weather systems, etc.).

Understanding differences and similarities within European airspace is crucial for mitigating risks and ensuring safe travel and efficient forecasting in different regions, particularly given the increasing turbulence features we reported over continental Europe. In light of our findings on the increasing occurrence of moderate-or-greater turbulence episodes in European airspace, there are significant economic implications for airlines. Indeed, while turbulence episodes rarely lead to crashes, they can damage cabin components such as seats and overhead bins, resulting from items falling or passengers colliding with them. These incidents not only necessitate repairs but also lead to flight delays and potential injuries to passengers and crew. Collectively, the expenses related to turbulence-induced damages, delays, and injuries can amount to up to 500 million euros for airlines each year (Sharman & Lane, 2016). With the increasing trend in turbulence episodes, airlines may face even greater financial burdens in the future, claiming for enhancing turbulence forecasting and mitigation strategies to minimize economic losses and ensure passenger safety, as well as, to improve routing and provide a way to make more sustainable aviation. Clearly, our study offers a starting point of using analog-based methodologies (Jézéquel, Dépoues, et al., 2018; Platzer et al., 2021; Yiou et al., 2017) for studying impacts of changing atmospheric circulation patterns on aviation turbulence, since data we used in this study representing turbulence in a parametric way (Hersbach et al., 2023). Further investigations will be devoted to study high-resolution resources (as meteorological satellites) over specific events and/or routes as well as convection-permitting numerical models to investigate both smaller spatial and projections to future climate scenarios (Coppola, Raffaele, et al., 2021; Coppola, Stocchi, et al., 2021).

Data Availability Statement

Data is available at Hersbach et al. (2023).

References

- Alberti, T., Anzidei, M., Faranda, D., Vecchio, A., Favaro, M., & Papa, A. (2023). Dynamical diagnostic of extreme events in Venice lagoon and their mitigation with the MoSE. Scientific Reports, 13(1), 10475. https://doi.org/10.1038/s41598-023-36816-8
- Bechtold, P., Bramberger, M., Dörnbrack, A., Leutbecher, L., & Isaksen, L. (2021). Forecasting clear-air turbulence. (Technical Memorandum No. 874). European Centre for Medium-Range Weather Forecasts (ECMWF).

Bevacqua, E., Maraun, D., Vousdoukas, M. I., Voukouvalas, E., Vrac, M., Mentaschi, L., & Widmann, M. (2019). Higher probability of compound flooding from precipitation and storm surge in Europe under anthropogenic climate change, Science Advances, 5(9), eaaw5531, https:// doi.org/10.1126/sciadv.aaw5531

Christidis, N., & Stott, P. A. (2021). The influence of anthropogenic climate change on wet and dry summers in Europe. Science Bulletin, 66(8), 813-823. https://doi.org/10.1016/j.scib.2021.01.020

Acknowledgments

TA and DF acknowledge useful discussions within the MedCyclones COST Action (CA19109) and the FutureMed COST Action (CA22162) communities. DF received support from the European Union's Horizon 2020 research and innovation programme under Grant agreement number 101003469 (XAIDA) and further support from the European Union's Horizon 2020 Marie Sklodowska-Curie Grant agreement number 956396 (EDIPI) and the LEFE-MANU-INSU-CNRS grant "CROIRE." We warmly thank five anonymous reviewers for fruitful suggestions helping us in significantly improving the results presented in this manuscript. Open access publishing facilitated by Istituto Nazionale di Geofisica e Vulcanologia, as part of the Wiley - CRUI-CARE agreement.

- Coppola, E., Raffaele, F., Giorgi, F., Giuliani, G., Xuejie, G., Ciarlo, J. M., et al. (2021a). Climate hazard indices projections based on CORDEX-CORE, CMIP5 and CMIP6 ensemble. *Climate Dynamics*, 57(5–6), 1293–1383. https://doi.org/10.1007/s00382-021-05640-z
- Coppola, E., Stocchi, P., Pichelli, E., Torres Alavez, J. A., Glazer, R., Giuliani, G., et al. (2021b). Non-hydrostatic RegCM4 (RegCM4-NH): Model description and case studies over multiple domains. *Geoscientific Model Development*, 14(12), 7705–7723. https://doi.org/10.5194/ gmd-14-7705-2021
- Dayan, U., Nissen, K., & Ulbrich, U. (2015). Review article: Atmospheric conditions inducing extreme precipitation over the eastern and western Mediterranean. Natural Hazards and Earth System Sciences, 15(11), 2525–2544. https://doi.org/10.5194/nhess-15-2525-2015
- Ellrod, G. P., & Knapp, D. I. (1992). An objective clear-air turbulence forecasting technique: Verification and operational use. *Weather and Forecasting*, 7(1), 150–165. https://doi.org/10.1175/1520-0434(1992)007(0150:AOCATF)2.0.CO;2
- Ellrod, G. P., & Knox, J. A. (2010). Improvements to an operational clear-air turbulence diagnostic index by addition of a divergence trend term. *Weather and Forecasting*, 25(2), 789–798. https://doi.org/10.1175/2009WAF2222290.1
- Faranda, D., Ginesta, M., Alberti, T., Coppola, E., & Anzidei, M. (2023). Attributing Venice Acqua Alta events to a changing climate and evaluating the efficacy of MoSE adaptation strategy. *npj Climate and Atmospheric Science*, 6(1), 181. https://doi.org/10.1038/s41612-023-00513-0
- Faranda, D., Messori, G., Coppola, E., Alberti, T., Vrac, M., Pons, F., et al. (2023a). Climameter: Contextualising extreme weather in a changing climate. EGUsphere, 2023, 1–34. https://doi.org/10.5194/egusphere-2023-2643
- Faranda, D., Messori, G., Jezequel, A., Vrac, M., & Yiou, P. (2023b). Atmospheric circulation compounds anthropogenic warming and impacts of climate extremes in Europe. *Proceedings of the National Academy of Sciences*, 120(13), e2214525120. https://doi.org/10.1073/pnas. 2214525120
- Faranda, D., Pascale, S., & Bulut, B. (2023). Persistent anticyclonic conditions and climate change exacerbated the exceptional 2022 European-Mediterranean drought. *Environmental Research Letters*, 18(3), 034030. https://doi.org/10.1088/1748-9326/acbc37
- Frehlich, R., & Sharman, R. (2004). Estimates of turbulence from numerical weather prediction model output with applications to turbulence diagnosis and data assimilation. *Monthly Weather Review*, 132(10), 2308–2324. https://doi.org/10.1175/1520-0493(2004)132(2308: EOTFNW)2.0.CO:2
- Gratton, G. B., Williams, P. D., Padhra, A., & Rapsomanikis, S. (2022). Reviewing the impacts of climate change on air transport operations. *Aeronautical Journal*, 126(1295), 209–221. https://doi.org/10.1017/aer.2021.109
- Gudmundsson, L., Seneviratne, S. I., & Zhang, X. (2017). Anthropogenic climate change detected in European renewable freshwater resources. *Nature Climate Change*, 7(11), 813–816. https://doi.org/10.1038/nclimate3416
- Gultepe, I., & Feltz, W. F. (2019). Aviation meteorology: Observations and models. Introduction. Pure and Applied Geophysics, 176(5), 1863–1867. https://doi.org/10.1007/s00024-019-02188-2
- Hersbach, H., Bell, B., Berrisford, P., Biavati, G., Horányi, A., Muñoz Sabater, J., et al. (2023). Era5 monthly averaged data on single levels from 1940 to present [Dataset]. Copernicus Climate Change Service (C3S) Climate Data Store (CDS). https://doi.org/10.24381/cds.f17050d7
- ICAO. (2018). Seventh meeting of the mog. Retrieved from https://www.icao.int/airnavigation/METP/Seventh%20Meeting%20of%20the% 20MOG/METPWGMOG-7_Attachment_SN_35_METP_4_WP.xxxx.Revised_EDR_values_Amd_79_vs_1.0_22mar2018.docx
- Jézéquel, A., Cattiaux, J., Naveau, P., Radanovics, S., Ribes, A., Vautard, R., et al. (2018a). Trends of atmospheric circulation during singular hot days in Europe. Environmental Research Letters, 13(5), 054007. https://doi.org/10.1088/1748-9326/aab5da
- Jézéquel, A., Dépoues, V., Guillemot, H., Trolliet, M., Vanderlinden, J.-P., & Yiou, P. (2018b). Behind the veil of extreme event attribution. *Climatic Change*, 149(3–4), 367–383. https://doi.org/10.1007/s10584-018-2252-9
- Kim, J.-H., & Chun, H.-Y. (2010). A numerical study of clear-air turbulence (cat) encounters over South Korea on 2 April 2007. Journal of Applied Meteorology and Climatology, 49(12), 2381–2403. https://doi.org/10.1175/2010JAMC2449.1
- Kim, J.-H., & Chun, H.-Y. (2011). Statistics and possible sources of aviation turbulence over South Korea. Journal of Applied Meteorology and Climatology, 50(2), 311–324. https://doi.org/10.1175/2010JAMC2492.1
- Kim, S.-H., Chun, H.-Y., Lee, D.-B., Kim, J.-H., & Sharman, R. D. (2021). Improving numerical weather prediction–based near-cloud aviation turbulence forecasts by diagnosing convective gravity wave breaking. Weather and Forecasting, 36(5), 1735–1757. https://doi.org/10.1175/ WAF-D-20-0213.1
- Kim, S.-H., Kim, J., Kim, J.-H., & Chun, H.-Y. (2022). Characteristics of the derived energy dissipation rate using the 1 Hz commercial aircraft quick access recorder (QAR) data. Atmospheric Measurement Techniques, 15(7), 2277–2298. https://doi.org/10.5194/amt-15-2277-2022
- Lane, T. P., Sharman, R. D., Trier, S. B., Fovell, R. G., & Williams, J. K. (2012). Recent advances in the understanding of near-cloud turbulence. Bulletin of the American Meteorological Society, 93(4), 499–515. https://doi.org/10.1175/BAMS-D-11-00062.1
- Lee, D. S., Fahey, D. W., Forster, P. M., Newton, P. J., Wit, R. C., Lim, L. L., et al. (2009). Aviation and global climate change in the 21st century. Atmospheric Environment, 43(22–23), 3520–3537. https://doi.org/10.1016/j.atmosenv.2009.04.024
- Lee, J. H., Kim, J.-H., Sharman, R. D., Kim, J., & Son, S.-W. (2023). Climatology of clear-air turbulence in upper troposphere and lower stratosphere in the northern hemisphere using era5 reanalysis data. *Journal of Geophysical Research: Atmospheres*, 128(1), e2022JD037679. https://doi.org/10.1029/2022JD037679
- Lee, S. H., Williams, P. D., & Frame, T. H. (2019). Increased shear in the north Atlantic upper-level jet stream over the past four decades. *Nature*, 572(7771), 639–642. https://doi.org/10.1038/s41586-019-1465-z
- Masson-Delmotte, V., Zhai, P., Pirani, A., Connors, S. L., Péan, C., Berger, S., et al. (2021). Climate change 2021: The physical science basis. Contribution of working group I to the sixth assessment report of the intergovernmental panel on climate change, 2.
- Naumann, G., Cammalleri, C., Mentaschi, L., & Feyen, L. (2021). Increased economic drought impacts in Europe with anthropogenic warming. *Nature Climate Change*, 11(6), 485–491. https://doi.org/10.1038/s41558-021-01044-3
- Ortiz-Bobea, A., Ault, T. R., Carrillo, C. M., Chambers, R. G., & Lobell, D. B. (2021). Anthropogenic climate change has slowed global agricultural productivity growth. *Nature Climate Change*, 11(4), 306–312. https://doi.org/10.1038/s41558-021-01000-1
- Otto, F. (2019). Attribution of extreme weather events: How does climate change affect weather? *Weather*, 74(9), 325–326. https://doi.org/10.1002/wea.3610
- Otto, F. E. L. (2016). Extreme events: The art of attribution. Nature Climate Change, 6(4), 342-343. https://doi.org/10.1038/nclimate2971
- Platzer, P., Yiou, P., Naveau, P., Filipot, J.-F., Thiébaut, M., & Tandeo, P. (2021). Probability distributions for analog-to-target distances. Journal of the Atmospheric Sciences, 78(10), 3317–3335. https://doi.org/10.1175/JAS-D-20-0382.1
- Prosser, M. C., Williams, P. D., Marlton, G. J., & Harrison, R. G. (2023). Evidence for large increases in clear-air turbulence over the past four decades. *Geophysical Research Letters*, 50(11), e2023GL103814. https://doi.org/10.1029/2023GL103814
- Rapella, L., Faranda, D., Gaetani, M., Drobinski, P., & Ginesta, M. (2023). Climate change on extreme winds already affects off-shore wind power availability in Europe. *Environmental Research Letters*, 18(3), 034040. https://doi.org/10.1088/1748-9326/acbdb2

- Scott, D., Gössling, S., & Hall, C. M. (2012). International tourism and climate change. WIREs Climate Change, 3(3), 213–232. https://doi.org/10. 1002/wcc.165
- Seneviratne, S. I., Zhang, X., Adnan, M., Badi, W., Dereczynski, C., Di Luca, A., et al. (2021). Weather and climate extreme events in a changing climate. In V. Masson-Delmotte (Ed.), et al. (Eds.), *Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change* (pp. 1513–1766). Cambridge University Press. https://doi.org/10. 1017/9781009157896.013
- Sharman, R., & Lane, T. (2016). Aviation turbulence (Vol. 10, pp. 978-983). Springer International Publishing.
- Shaw, T. A., & Miyawaki, O. (2023). Fast upper-level jet stream winds get faster under climate change. Nature Climate Change, 14(1), 61–67. https://doi.org/10.1038/s41558-023-01884-1
- Smith, I. H., Williams, P. D., & Schiemann, R. (2023). Clear-air turbulence trends over the north Atlantic in high-resolution climate models. *Climate Dynamics*, 61(7–8), 1–17. https://doi.org/10.1007/s00382-023-06694-x
- Storer, L. N., Williams, P. D., & Gill, P. G. (2019). Aviation turbulence: Dynamics, forecasting, and response to climate change. Pure and Applied Geophysics, 176(5), 2081–2095. https://doi.org/10.1007/s00024-018-1822-0
- Storer, L. N., Williams, P. D., & Joshi, M. M. (2017). Global response of clear-air turbulence to climate change. Geophysical Research Letters, 44(19), 9976–9984. https://doi.org/10.1002/2017gl074618
- Trier, S. B., Sharman, R. D., Muñoz-Esparza, D., & Keller, T. L. (2022). Effects of distant organized convection on forecasts of widespread clearair turbulence. *Monthly Weather Review*, 150(10), 2593–2615. https://doi.org/10.1175/MWR-D-22-0077.1
- Williams, P. D. (2017). Increased light, moderate, and severe clear-air turbulence in response to climate change. Advances in Atmospheric Sciences, 34(5), 576–586. https://doi.org/10.1007/s00376-017-6268-2
- Williams, P. D., & Joshi, M. M. (2013). Intensification of winter transatlantic aviation turbulence in response to climate change. Nature Climate Change, 3(7), 644–648. https://doi.org/10.1038/nclimate1866
- Williams, P. D., & Storer, L. N. (2022). Can a climate model successfully diagnose clear-air turbulence and its response to climate change? Quarterly Journal of the Royal Meteorological Society, 148(744), 1424–1438. https://doi.org/10.1002/qj.4270
- Wolff, J., & Sharman, R. (2008). Climatology of upper-level turbulence over the contiguous United States. Journal of Applied Meteorology and Climatology, 47(8), 2198–2214. https://doi.org/10.1175/2008jamc1799
- Yiou, P., Jézéquel, A., Naveau, P., Otto, F. E. L., Vautard, R., & Vrac, M. (2017). A statistical framework for conditional extreme event attribution. Advances in Statistical Climatology, Meteorology and Oceanography, 3(1), 17–31. https://doi.org/10.5194/ascmo-3-17-2017